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Not Activated

TBA2

_

Not Activated

TBB1

_

Not Activated

TBB2

_

Not Activated

TBC1

MBAL2066

Connection

300 K

TBC2

SSS002

EX1 Extra Test (next SSS13)

300 K

TBD1

MBAL3060

Disconnection

300 K

TBD2

MBAR3010

Prep 5 Pumping + Cool Down

Cooling

TBE1

MBAL1066

Prep 5 Pumping + Cool Down

Cooling

TBE2

MBBL 3092

Connection

300 K

TBF1

MBAL3056

Cold test

1.9 K

TBF2

MBAL2069

Cold test (end)

1.9 K

SM18 Operation SCHEDULE Meeting (Tuesdays)

 Bench status  and tests for this week : 

· Shafts in 2069 (F1-QLOC); 3010 (TBD2-MM); 2066 (TBC1-MM).

Test Coordinator this week:  Nicolay (164674); MTM support on call Guy:  (16-3602).

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

1. Leaks: 3105 (leak in the magnet); 3056 (joint in the flange of the long anticryostat).
2. Process: Lost of cryo OK in TBC2: Level in pressure detectors  changed. Solved. Lost of cryo OK after small quenches : level in the probe in the current leads corrected.
· Other General Issues
1. Request of F.Flamand : After a MM, please plug the cover of the connector cables coming from the shaft.

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Thursday 11_A

Friday 12_M


	TBD2: The shaft could not rotate well in aperture 1 when it is closed to the MRB. 

TBD2: Rotation in aperture 1 not good (3%). The “flat cables” is scratching the the protection cover.

LHC cycle: Over-ranges in aperture 1 at high currents: 
	ICS
	
	Pierre, Stephane informed. Common decision was to use as it is because the situation  looks like better in MM and quench position.

Origin of problem found by mecanician team . J.Halik will inform ICS

Maryline informed Stephane. OK to continue.

Maryline informed Stephane.

He advises to reduce the speed for the load line.

	Thursday 11_A


	TBD2: Crate 2 slot 8 Tcoil out of range.

Investigation by Maryline :ADC card not working.. 
	AB-CO
	
	Card changed Monday.

	Tuesday 09_M

Friday 12_ M


	All benches with MM : Problem in Sun MTA24. Display is  not working. Vi missing

Problem appeared again.
	AB-CO
	
	Repaired by A.Raimondo.

Investigation on the run.

	Thursday 11_A, N


	Cluster E, D,F: HWREC switched off because MTF is not running (upgrade) .

When MTF operational : no access to the DB for cluster D, E.

Quit all application, restart TEMA, Re-launched again : Error was : error updating DB unable to access hr_chklst”


	TS/CSE


	MTF

HWREC
	.

Marlyne’s  instructions:HWREC stopped again . Names entered manually in TEMA for cluster D and E.

Maryline generated the settings manually.

Maryline informed T.Ladzinski.

He solved the problem. An immediate check is now added in the procedure after each upgrade.

	Thursday 11_N


	TBD2: SPA test. Interlock water. Electro valve was closed (low water pressure). 

TEMA was launched but hf application hung.


	TS/EL

AB-CO
	TEMA
	Other valves closed and interlock disappears (Florence).

Re-launch and OK (Florence)

	Friday 12_M


	TBE1: Task phase not signed (break vacuum) because L.Herblin asked to make a leak test before. ICS activity stopped. 

J.Axensalva contacted to change the status: stripping to stand by.
	
	
	

	Friday 12_A


	TBC1, TBD2: HV tests: Some new serial ports installed by AB-Co in fixed rack are not working properly.


	AB-CO
	
	Richard consulted. 

HV Tests performed using other serial ports.

M.Duron is investigating to solve the problem (Monday).



	Friday 12_A


	TBE1: Problem in the Task Tracking System. ICS did not see the changes in the task because there is no automatic refresh,


	ACR
	
	J.Axensalva is solving this problem.



	Sat 13_N


	TBD2: HV initial at warm . The meeger stopped four times during the test. Run-recovery then “error 15530”


	ACR
	HV
	OK after shaking ??? communication cables.

(Jackie)



	Sunday 14_N


	TBD2: Alarm anti-cryostat during the night.


	MTM
	
	Support not called ??










